Author's response to reviews

Title: Effects of acupuncture treatment on depression insomnia: A study protocol of a multi-centre randomized controlled trial

Authors:

Yuan-Fang Chen (yfchen1@126.com)
Zhao-Hui Liang (liang_zhao_hui@163.com)
Neng-Gui Xu (ngxu8018@gzhtcm.edu.cn)
Zhen-Hua Xu (xzh197011@163.com)
Jian-Hua Liu (ljh1008@21cn.com)
Wen-Bin Fu (fuwenbin@139.com)
Shu-jun Xu (13826497503@163.com)

Version: 8 Date: 6 December 2012

Author's response to reviews:

1 About the language, we improved the article and one of the native speaker read straight through it and the comments are applied.

2 Some detail has been added to “the sample size calculation”.

3 “The third-party assessment” was part of “blind method”, this part has been rewritten as a separate section which has been highlighted.